Author's response to reviews

Title: Evaluation of pain and function after two home exercise programs: A randomized controlled trial on women with chronic neck pain

Authors:

Linn Karlsson (linn.karlsson@liu.se)
Esa-Pekka Takala (Esa-Pekka.Takala@ttl.fi)
Bjorn Gerdle (bjorn.gerdle@liu.se)
Britt Larsson (britt.larsson@liu.se)

Version: 2 Date: 24 June 2013

Author's response to reviews:

Dear Editor-in-Chief,

Please find the submitted manuscript “Evaluation of pain and function after two home exercise programs: A randomized controlled trial on women with chronic neck pain” to be considered for publication in BMC Musculoskeletal Disorders.

Pre-review revisions has been made:

1) The last section of the abstract is corrected. Trial registration has been included, with trial registration number, marked with yellow.

2) Legends to the figures has been removed in the separate figure files.

Yours sincerely,

Linn Karlsson, PT, PhD student
Rehabilitation Medicine,
Dept. of Medicine and Health Sciences (IMH)
Faculty of Health Sciences
University of Linköping
SE-581 85 Linköping
Sweden
Telephone: +46 101034933
E-mail: linn.karlsson@liu.se